Author's response to reviews

Title: Revisiting the technical validation of tumour biomarker assays: How to open a Pandora's box

Authors:

Caterina Marchiò (caterina.marchio@unito.it)
Mitch Dowsett (Mitch.Dowsett@icr.ac.uk)
Jorge S Reis-Filho (Jorge.Reis-Filho@icr.ac.uk)

Version: 5 Date: 23 March 2011

Author's response to reviews:

Dear Sirs,

we have re-formatted the manuscript as requested and the article conforms now to the journal style.
In particular:
- the introduction is now a subheading and a section of the “Review”;
- the structure of the manuscript is now in accordance with the journal style (Abstract, Review, Conclusions, Abbreviations; Competing interests; Authors' contributions; Acknowledgements; References)
- we have checked our manuscript for typographical errors that have been corrected.

All authors have read and approved the revised version of the manuscript.
Should you have any queries, please do not hesitate to contact us.
Many thanks for this great opportunity.
I look forward to hearing from you in due course.

Best wishes,
Caterina Marchiò